

# **Geophysical Research Letters**<sup>\*</sup>

# **RESEARCH LETTER**

10.1029/2022GL102485

#### **Key Points:**

- We exploit D-amino acid biomarkers in multi-year deep-sea sediment trap time series to evaluate bacterial contribution to total export
- Bacterial detritus accounts for up to 19 ± 8% of sinking POC and up to 36 ± 14% of PN, making up a large unrecognized part of biological pump
- The relative contribution of bacterial detritus to sinking particles increases with decreased export production

#### Supporting Information:

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

Y. Shen, yuanshen@xmu.edu.cn

#### Citation:

Shen, Y., Guilderson, T. P., Chavez, F. P., & McCarthy, M. D. (2023). Important contribution of bacterial carbon and nitrogen to sinking particle export. *Geophysical Research Letters*, 50, e2022GL102485. https://doi. org/10.1029/2022GL102485

Received 8 DEC 2022 Accepted 26 MAY 2023 Corrected 3 JUL 2023

This article was corrected on 3 JUL 2023. See the end of the full text for details.

© 2023. The Authors. This is an open access article under

the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# Important Contribution of Bacterial Carbon and Nitrogen to Sinking Particle Export

Yuan Shen<sup>1</sup>, Thomas P. Guilderson<sup>2,3</sup>, Francisco P. Chavez<sup>4</sup>, and Matthew D. McCarthy<sup>2</sup>

<sup>1</sup>State Key Laboratory for Marine Environmental Science, College of Ocean and Earth Sciences, Xiamen University, Xiamen, China, <sup>2</sup>Ocean Sciences Department, University of California, Santa Cruz, CA, USA, <sup>3</sup>Lawrence Livermore National Laboratory, Livermore, CA, USA, <sup>4</sup>Monterey Bay Aquarium Research Institute, Moss Landing, CA, USA

**Abstract** Photosynthesis in the surface ocean converts atmospheric  $CO_2$  into organic particles, with the fraction sinking to depth representing a major part of the ocean's biological pump. Although sinking particles are known to be altered by attached-bacteria during transit, most prior organic geochemical data indicated only minor replacement of plankton-derived particles by bacterial material. We exploit bacteria-specific biomarkers (D-amino acids) in a multi-year sediment trap in the Pacific Ocean (1,200 m) and suggest a different view. Major D-amino acids were consistently measured at abundance demonstrating widespread accumulation of bacterial material in sinking particles. Bacterial detritus was estimated to account for up to 19% of particulate organic carbon and up to 36% of particulate nitrogen, much higher than cell count-based values. The bacterial relative contribution increased with decreasing export production. Our results indicate that bacterial material constitutes an underappreciated component of the biological pump, a role expected to rise as the ocean warms.

**Plain Language Summary** Phytoplankton photosynthesis in the surface ocean plays a critical role in stabilizing atmospheric  $CO_2$ . It converts  $CO_2$  into organic particles that sink and are reworked by colonizing bacteria. Bacteria respire most particles back to  $CO_2$  while also transforming some into their cell components. Although the involvement of bacteria can replace the plankton-derived particles to bacterial material, most past organic geochemical data have suggested that the deep-sea particles are still comprised mainly of plankton remnants. This renders the contribution of bacterial material to total particle export an unresolved and yet important question, because the source and composition of particles are important to their fate in the ocean. Here, we analyzed bacteria-specific molecules in deep-sea sinking particles and found that bacterial organic matter actually made up a large fraction of the particles. In addition, the relative contribution of bacterial material to the sinking particles increased as the total carbon export decreased. This has important implications for the future ocean carbon cycle, because modeling work predicts a scenario of lower carbon export to the deep sea as the ocean warms. In this context, our findings imply a greater importance of bacteria in marine organic matter export and sequestration in a warming ocean.

## 1. Introduction

A fraction of fixed carbon in the surface ocean sinks to depth as detrital particles containing a mixture of phytodetritus, consumer fecal pellets, and other organic aggregates (Turner, 2015). The downward transit and remineralization of this particulate material play an important role in stabilizing atmospheric carbon dioxide and supporting deep-sea metabolism, together representing a main component of the ocean's biological pump (Arístegui et al., 2009; Herndl & Reinthaler, 2013). The large majority of sinking particulate organic matter (POM) is respired and altered by bacteria during transit to the sea floor (Bidle & Azam, 1999; Smith et al., 1992). The extent to which the particles are altered exerts large scale feedback on global warming and is directly linked to particle source and composition.

Past data based on molecular, isotopic, spectroscopic, and genomic approaches offered varied perceptions on the composition of the deeply exported particles (Wakeham & Lee, 2019). Chemical structures resolved by solid-state <sup>13</sup>C NMR and pyrolysis mass spectrometry showed a remarkable compositional similarity between the deeply exported particles and surface phytoplankton, suggesting non-selective preservation of plankton material in sinking particles (Hedges et al., 2001; Minor et al., 2003). Additional insights from the Redfield stoichiometry, confocal microscopy, bulk stable and radioactive isotopes, as well as the emerging compound-specific amino acid stable isotopes have indicated a relatively "fresh" planktonic signature in midwater and deep-sea sinking particles

(Hernes et al., 2001; McCarthy et al., 2007; Robinson et al., 2012; Roland et al., 2008; Shen et al., 2021; Waite et al., 2000). Together, much of this past work has created a paradigm that despite the large degree of bacterial remineralization, the deeply exported particles are composed predominately of leftovers of surface plankton. However, at the same time molecular level characterizations based on conventional hydrolysis and chromatograph approaches have shown a rapid depletion in common biochemicals with depth (e.g., amino acids, carbohydrates, lipids), with varying degrees of changes in molecular composition (Hernes et al., 1996; Lee et al., 2000; Wakeham et al., 1997). These data suggest substantial microbial removal of algal OM along with partial replacement by bacterial remnants. Recent studies based on genomic approaches have also pointed to large presence and activity of diverse groups of colonizing bacteria in the ocean (Boeuf et al., 2019; Johnson et al., 2020; Mestre et al., 2018; Preston et al., 2020; Zhao et al., 2020), implying accumulation of bacterial biomass in particles.

A holistic interpretation of these seemingly inconsistent data types has been unclear, due mainly to the lack of studies that directly quantify the extent of bacterial material input to sinking POM. Early work attempting to estimate bacterial contribution in organic carbon pool has focused mainly on non-sinking POM. Based on measurements of bacterial cell number, a large and variable contribution of bacterial organic carbon to total particulate organic carbon (from <13% to 70%) has been estimated in the upper water column of northeast Pacific Ocean and central north Pacific gyre (Azam et al., 1992; Cho & Azam, 1988; Simon et al., 1990). For sinking POM, the few existing quantitative estimates of bacterial biomass were also based on bacterial cell counts and they indicated only a very small bacterial carbon contribution (1% to  $\sim$ 4%) to the sinking organic carbon in the mesopelagic and deep ocean (Ducklow et al., 1985; Turley & Mackie, 1994). The substantially low bacterial fraction estimated for sinking POM imply that the cell count-based approach may have underestimated bacterial material entrained in sinking particles. Given that the vast majority of organic matter in the ocean is present as nonliving forms (Benner & Kaiser, 2003; Hedges & Keil, 1995), estimates based on intact cell counts would likely provide a lower bound on actual export of bacterial material to the deep sea.

To obtain more accurate estimates of the bacterial material contributions to sinking POM, specific organic tracers for bacteria detritus are needed. Bacteria synthesize a variety of unique molecules ("biomarkers") that can trace the occurrence of bacterially-produced material in natural organic matter mixtures. Such biomarkers include D-enantiomers of amino acids, muramic acid, diaminopimelic acid, and short-chain 3-hydroxy fatty acids (Kaiser & Benner, 2008; Kawasaki & Benner, 2006; Schleifer & Kandler, 1972; Wakeham et al., 2003). Among these, the D-amino acids have been the most extensively studied, and are perhaps the most representative biomarkers of bacterial organic matter given their high abundance and widespread occurrence in diverse bacterial macromolecules (e.g., peptidoglycan, lipopeptides, etc.) (Schleifer & Kandler, 1972). D-amino acids have also been developed as quantitative tracers for estimating the relative bacterial contribution to marine dissolved organic matter (DOM) and suspended POM (Kaiser & Benner, 2008; Kawasaki et al., 2011; Tremblay et al., 2015). Despite the uncertainties associated with the potentially dissimilar reactivities among organic components (e.g., peptidoglycan degrades slower than protein; Nagata et al., 2003), the D-amino acids-based estimates represent one of the few available approaches to quantify bacterial material in natural organic matter. To our knowledge, no published studies have examined D-amino acids in sinking POM.

In this study, we present what we believe to be the first D-amino acid measurements in a long-term sediment trap time series deployed at 1,200 m depth on the California margin. We use these data to evaluate the abundance of bacterial material in deep-sea sinking particles, explore the temporal trends of bacterial material export, and quantify the contribution of bacterial organic carbon and nitrogen to the total sinking particle fluxes. Further, we discuss the implications for carbon export and sequestration in the context of a warming ocean.

# 2. Materials and Methods

The study region, Monterey Bay, is situated on the central coast of California (Figure S1 in Supporting Information S1) and characterized by a strong seasonality in hydrographic conditions and biological production (Chavez et al., 2017). These factors provide an ideal setting to evaluate bacterial signatures in exported particles under varying conditions (see more information in Supporting Information S1). Sinking particle samples were collected between August 1998 and December 2004 using a Honjo Mark IV sediment trap as described previously (Castro et al., 2018; Shen et al., 2021). The trap was deployed at the M2 mooring station (36.697°N, 122.378°W; Figure S1 in Supporting Information S1) at 1,200 m depth and was pre-programmed to collect sinking particles every 14 days. Samples recovered from the field were wet-sieved (1 mm), rinsed with distilled water, centrifuged, and then dried at 60°C for 24 hr. Bacterial alteration of the particle samples during the long-term storage was not evident from analytical data in both this and another associated project (see discussion in Supporting Information S1). Approximately 20 mg of dried sample was weighed for measurements of particulate organic carbon (POC) and total particulate nitrogen (PN) using a Carlo-Erba NA-1100 analyzer. Samples for the determination of POC were decarbonated with addition of 15% hydrochloric acid (HCl) prior to analysis. The actual weights of samples were recorded and used to calculate the fluxes of POC and PN.

D- and L-enantiomers of amino acids were determined using an Agilent 1,260 ultrahigh performance liquid chromatography (HPLC) system equipped with a Poroshell 120 EC-C18 column and a fluorescence detector. Approximately 100–200 µg of dried sample (weight recorded) was hydrolyzed in sealed glass vials with 6 mol L<sup>-1</sup> HCl at 110°C for 20 hr to liberate amino acids. The hydrolyzed samples were dried with N<sub>2</sub> and immediately derivatized with *o*-phthaldialdehyde and *N*-isobutyryl-*L*-cysteine (Shen et al., 2017). A total of eighteen (18) hydrolyzable amino acids were separated, including the non-protein β-alanine and γ-aminobutyric acid. The four D-amino acids commonly found in marine organic matter were reported in this study: D-Asx (asparagine + aspartic acid), D-Glx (glutamine + glutamic acid), D-Ser (serine), and D-Ala (alanine). Acid-catalyzed racemization was corrected according to Kaiser and Benner (2005). The amount of D-amino acids determined was divided by total POC or PN to calculate D-amino acid yield value. This yield value was multiplied by POC or PN flux to calculate the flux of D-amino acids (in units of µmol AA-C m<sup>-2</sup> d<sup>-1</sup> or µmol AA-N m<sup>-2</sup> d<sup>-1</sup>).

The relative contribution (%) of bacterial carbon and nitrogen to the total POC and PN was estimated from p-amino acids, following the formula and endmembers of Kaiser and Benner (2008):

Bacterial C or N contribution (%) = D-amino acid yield<sub>sample</sub>/D-amino acid yield<sub>bacteria</sub>  $\times$  100,

where D-amino acid yield<sub>sample</sub> and D-amino acid yield<sub>bacteria</sub> are the POC- and PN-normalized values of two D-amino acids (i.e., D-Glx and D-Ala following the choice of Kaiser & Benner, 2008) in sediment trap material and cultured marine bacterial cell, respectively. The D-amino acid yields are reported with units of  $\mu$ mol AA-C mmolPOC<sup>-1</sup> or  $\mu$ mol AA-N mmolPN<sup>-1</sup>. The endmember values for D-amino acid yield <sub>bacteria</sub> were obtained from a mixture of 20% autotrophic bacteria and 80% heterotrophic bacteria (Kaiser & Benner, 2008), averaged at 4.7  $\mu$ mol AA-C mmolPOC<sup>-1</sup> and 6.0  $\mu$ mol AA-N mmolPN<sup>-1</sup>.

# 3. Results and Discussion

## 3.1. Flux and Export Pattern of Total Bacterial Material to the Deep Ocean

Sinking particle fluxes measured at 1,200 m of Monterey Bay showed large temporal changes (Figures 1a and 1b; Table S1 in Supporting Information S1), reflecting the pronounced variation in overlying plankton production. Mass fluxes of POC and the total of 18 hydrolyzable amino acids (THAA) during 1998–2004 varied by over an order of magnitude from 1.2 to 11.8 mmol C m<sup>-2</sup> d<sup>-1</sup> and from 91 to 1,305 µmol AA-C m<sup>-2</sup> d<sup>-1</sup>, respectively. Fluxes of THAA closely followed those of POC (Pearson correlation: r = 0.84, p < 0.001) and also generally increased during the more productive months (April–August). Fluxes of THAA determined in this study were overall higher and more variable than those measured previously in open ocean deep water (>1,000 m), including the equatorial Pacific Ocean (15–300 µmol AA-C m<sup>-2</sup> d<sup>-1</sup>; Gupta & Kawahata, 2002; Lee et al., 2000), north subtropical Pacific Ocean (<1 µmol AA-C m<sup>-2</sup> d<sup>-1</sup>; Sargasso Sea: 4–55 µmol AA-C m<sup>-2</sup> d<sup>-1</sup>; Ittekkot et al., 1984; Salter et al., 2010).

On average, the THAA-carbon accounted for  $9 \pm 3\%$  (range: 4%–13%) of the total POC (Table S1 in Supporting Information S1). This value was substantially lower than those commonly observed in freshly-produced marine POM (e.g., >50% of POC in the upper equatorial Pacific Ocean), but was ~10-fold higher than that of its dissolved counterparts (i.e., 0.5%–1.0% of DOC in the deep Pacific and Atlantic Oceans; Kaiser & Benner, 2009; Lee et al., 2004). While THAA concentrations in the DOM and POM pools may not be exactly comparable, the relative content of THAA (%OC) has been widely demonstrated to be a fairly universal proxy for relative lability in marine DOM, POM, and sediments (Benner & Amon, 2015; Cowie & Hedges, 1994; Lee et al., 2000). The observed moderate relative abundance of THAA in this study suggests that the particles exported to deep water have been substantially altered by degradation, but were still of relatively high lability compared to ambient deep-sea DOM.





**Figure 1.** Changes in mass fluxes of (a, b) total particulate organic carbon and total hydrolyzable amino acids (THAA) and (c, d) four major D-amino acids as a function of collection date (shown as the medium date of each deployment). D-Asx: D-asparagine + D-aspartic acid; D-Glx: D-glutamine + D-glutamic acid; D-Ser: D-serine; D-Ala: D-alanine. Upwelling months (March–August) are marked with light blue bands. Errors were determined by propagating uncertainties from amino acid instrumental measurements (the Relative Standard Deviation of replicate injections for single amino acid, i.e., RSD is ~1%) and racemization corrections (Kaiser & Benner, 2005).

The four D-amino acid biomarkers (D-Asx, D-Glx, D-Ser, D-Ala) unique to bacteria were consistently measured in all samples (Figures 1c and 1d; Table S1 in Supporting Information S1), indicating a widespread bacterial imprint of heterotrophic or/and autotrophic bacteria. Mass flux of individual D-amino acids at 1,200 m varied substantially across the 6-yr sampling period (Figures 1c and 1d). The sum of the four D-amino acid fluxes varied from 1.2 to 10.1 µmol AA-C m<sup>-2</sup> d<sup>-1</sup> with a flux-weighted average of  $5.5 \pm 2.4$  µmol AA-C m<sup>-2</sup> d<sup>-1</sup> (Table S1 in Supporting Information S1). For context, this D-amino acid flux is not trivial, with its magnitude comparable to that of total THAA fluxes measured previously in the deep subtropical Pacific and Atlantic Oceans (<1–55 µmol AA-C m<sup>-2</sup> d<sup>-1</sup>; Ittekkot et al., 1984; Wakeham & Lee, 1989). Given that D-amino acids by themselves represent only one kind and also minor component of the many compound classes (e.g., peptidoglycan, lipopeptides, etc.) that constitute the bacterial cells, the observed large flux of D-amino acids alone implies that the export of total bacterial detritus to the mesopelagic and deep ocean is substantial.

Little is known about the temporal export pattern of bacterial material in the ocean. Careful examination of the relationships between D-amino acid flux and the total POC flux revealed a somewhat surprising, but highly consistent, parabolic pattern (quadratic regression:  $r^2 = 0.50-0.78$ , p < 0.001; Figure 2). Bacterial material flux, as indicated by D-amino acid flux, was low when POC flux was the lowest or highest. The highest bacterial material flux was observed when POC flux was intermediate in value (Figure 2). These results suggest that the mass flux of bacterial material is highly variable and is not a simple linear function of the total export production. One major implication is that it is not possible to directly estimate the flux of bacterial material using a single fixed percentage of total export production.

This unexpected export pattern could be explained by the variable relative proportion of bacterial material in bulk particles that ultimately leads to the observed parabolic pattern of bacterial material flux (Figure S2 in Supporting Information S1). The relative proportion of bacterial material in sinking particles varied in an opposite direction to the total particle flux (Figure S3 in Supporting Information S1), a feature that appears to be related to particle





**Figure 2.** Changes in D-amino acid mass flux as a function of particulate organic carbon (POC) flux. The black solid curves represent the quadratic polynomial fits. Blue circles: samples collected during fall and winter; Orange circles: samples collected during spring and summer. Errors were determined in the same way as those in Figure 1.

properties and overlying plankton dynamics. Lower primary productivity periods dominated by small phytoplankton (e.g., cyanobacteria) tend to produce particles with smaller size, higher porosity and lower sinking rate, which extends the transit time of particles and presumably facilitates heterotrophic bacterial growth and transformation (Goutx et al., 2007; McDonnell et al., 2015; Spilling et al., 2023). Occurrence of cyanobacteria may themselves also contribute D-amino acids (D-Glx, D-Ala). In contrast, higher primary production periods dominated by large phytoplankton tend to produce larger particles with smaller relative bacterial contributions (Figure S4 in Supporting Information S1).

#### 3.2. Contribution of Bacterial Material to Marine Particle Export

We compared D-amino acids yields with values measured previously in pure marine bacterial cultures to evaluate the relative contribution (%) of total bacterial material to sinking particles. Estimation based on this approach in principle includes both autotrophic and heterotrophic bacteria, but as discussed below, multiple lines of evidence indicate heterotrophic bacteria as the dominant source. The calculations suggested that bacterially-derived organic carbon and nitrogen accounted for  $4 \pm 2\%$  to  $19 \pm 8\%$  of the total sinking POC and  $7 \pm 3\%$  to  $36 \pm 14\%$ of total PN, respectively, with the bacterial contribution increasing with declining POC and PN fluxes (Figure 3). As discussed below, these estimates likely represent an upper bound on the bacterial contribution.

These first geochemical estimates of total bacterial organic matter in sinking POM are far higher than previous estimates based on cell counts, and yet seem to fall within a reasonable range. Regressing %bacterial contribution to the total POC and PN fluxes returned relatively high y-intercept values (C: ~16%; N: ~31%) at zero flux (Figure 3). The intercepts predict the theoretical average %bacterial contribution for scenarios of no net export of sinking organic matter. These theoretical values were comparable to those estimated for non-sinking marine



**Figure 3.** Estimated contributions (%) of bacterial carbon and nitrogen to the total sinking particulate organic carbon (POC) and particulate nitrogen (PN) and their relationships with POC and PN fluxes. Errors were determined by propagating uncertainties from the following sources: D-amino acid instrumental measurements (RSD  $\sim$ 1%), racemization corrections (Kaiser & Benner, 2005), and bacterial D-amino acid endmembers (Kaiser & Benner, 2008).

DOM and suspended POM (C:  $\sim 25\%$ , N:  $\sim 50\%$ ; Kaiser & Benner, 2008; Kawasaki et al., 2011; Tremblay et al., 2015). Importantly, our values were about one order of magnitude higher than the previous cell-count-dependent estimates in sinking POC (1%–4%; Ducklow et al., 1985; Turley & Mackie, 1994). Our estimates suggest that bacterial detritus represents a large fraction of the deeply exported organic matter, substantially beyond that indicated by prior quantitative estimates.

This biomarker-based approach has been broadly applied in organic geochemical literature (Kaiser & Benner, 2008; Kawasaki et al., 2011; Lehmann et al., 2020; Tremblay et al., 2015). However, it is important to note that there are important uncertainties associated with such estimates. It first assumes that D-amino acids track all of bacterial biomass including cell walls and intracellular components. D-amino acids are known to occur in various components of the bacterial cell wall (e.g., peptidoglycan, siderophores, lipopolysaccharides, lipopeptides; Kaiser & Benner, 2008; Schleifer & Kandler, 1972), and thus would most directly trace cell wall components. Previous experiments measuring the hydrolysis and remineralization rates of bacterial cell wall components have suggested that peptidoglycan is more resistant to biodegradation than protein (Nagata et al., 2003). As such, endmember values of *D*-amino acid yields determined in bacterial cultures (mostly whole cells) are likely lower than those in detrital cells. Therefore, our estimates based on fresh cells might provide an upper bound on the bacterial contribution. A second assumption is that bacterial species used to establish endmembers match those present in the field. This is currently difficult to assess because marine natural bacterial species are poorly constrained, and the majority cannot be cultured. Nevertheless, nearly all the bacterial species used as endmembers for this study are found in the study region (e.g., autotrophs: Synechococcus sp., Trichodesmium; heterotrophs: Alteromonas sp., Cytophaga sp., Roseobacter litoralis, Pseudomonas sp.; Farnelid et al., 2016; Suzuki et al., 2001). The variants of these same caveats, while not aways directly discussed, are inherent in essentially all organic biomarker-based estimations about source of detrital organic matter.

Regarding the specific sources of D-amino acids measured in sinking particles, they could be derived from heterotrophic bacteria or/and autotrophic bacteria. The two D-amino acids that are unique to heterotrophic bacteria (D-Asx and D-Ser) were consistently measured (Figure 1). The D/L ratios of Asx and Ser were many-fold higher than the levels found in fresh heterotrophic bacterial POM (Figure S5 in Supporting Information S1), indicating extensive heterotrophic degradation which leads to preferential removal of L-amino acids over that of D-amino acids (see Section IV in Supporting Information S1). Overall, the specific D/L amino acid signatures suggest that the majority of bacterial material observed in sinking particles was most likely derived from heterotrophic bacteria.

There are multiple possible sources for the heterotrophic bacterial material detected in sinking particles, including heterotrophic bacteria that reside in the surface ocean, bacterial growth on particles during sinking, bacteria from zooplankton guts, or aggregation of bacterial detritus. The relative contribution of these sources is difficult to resolve with our current data set. Nevertheless, of these, bacteria from guts of grazers are likely not a major contributor, as suggested by the paired analysis of trophic position (as a rough proxy of the contribution of grazers; Shen et al., 2021) that showed almost no relationship between D-amino acid export and particle trophic position (Figure S6 in Supporting Information S1). Based on recent studies, we propose that the most likely source is heterotrophic bacteria "growing in" and transforming plankton-derived particles into bacterial material during the long transit (weeks to months), producing most of the bacterial organic matter detected. This might represent the most parsimonious interpretation in light of recent work that reveal rapid bacterial growth and production upon addition of sinking particles (Church et al., 2021), predominant particle-associated lifestyle of deep-sea microorganisms (Zhao et al., 2020), and intimate relationships between bacterial community dynamics (e.g., richness, population density) and rates of particle remineralization (Baumas et al., 2021; Nguyen et al., 2022). Incubation studies in both DOM and POM have shown that fresh algal organic matter is remarkably degraded on short timescales, with bacteria rapidly consuming eukaryotic algal material and producing refractory bacterial material on a scale of days to weeks (Calleja et al., 2013; Lehmann et al., 2020; Shen & Benner, 2020). Finally, the inverse relationships observed between D-amino acid yields (as a proxy of relative bacterial contribution) and surface water Chl-a and phytoplankton biomass in the study region are also consistent with the expectation of more bacterial material "growing" in with slower sinking rates (see Figure S4 and Text in Supporting Information S1).

#### 3.3. Implications for Carbon Export and Sequestration in the Deep Ocean

An outstanding observation in this study is the widespread imprint of bacterial material in sinking particles, with nearly one order of magnitude greater bacterial contribution being suggested compared to the past cell count-based estimates. These findings suggest that the bacterially-derived material play a far more important role in marine carbon export and sequestration than past organic geochemical data would suggest. Importantly, residual bacterial organic matter is generally found to be more resistant to degradation than planktonic organic matter (Amon et al., 2001; Lehmann et al., 2020; Nagata et al., 2003). Therefore, while heterotrophic bacterial biomass synthesized in sinking particles is ultimately derived from surface phytoplankton, the bacterial activity on particles could represent an important unappreciated factor in carbon sequestration by converting labile phytoplankton carbon into bioresistant bacterial carbon. This process conceptually resembles that of microbial carbon pump (MCP), which suggests bacterial growth in the water column can rapidly transform labile substrates into refractory dissolved molecules that contribute to the millennia-year old DOC pool (Jiao et al., 2010; Lechtenfeld et al., 2015; Ogawa et al., 2001). The MCP concept has up to now only been used for the DOM pool and free-living bacteria, and we suggest that similar carbon sequestration processes linked to heterotrophic transformation could also occur in the POM pool. This idea is supported by the remarkable similarity in D/L distribution patterns between sinking particles (i.e., D/L ratio values: Asx > Glx  $\approx$  Ser < Ala) and those observed in suspended POM, as well as DOM from various oceanic regions (Figure S7 in Supporting Information S1).

Building upon the present work on POM and prior studies on DOM, we propose that bacteria colonizing on particles have the potential to mediate at least three different pathways that assist carbon sequestration in the ocean (Figure 4). The first is via the transformation of plankton-derived organic matter into bacterial cell components with reduced reactivity (e.g., peptidoglycan; Nagata et al., 2003). Data from this study suggest that this pathway in fact represents an appreciable fraction of the total carbon export, which has not previously been widely recognized. A second pathway is through the extracellular release of bioresistant dissolved molecules to ambient water during the growth of particle-attached bacteria. Previous bioassay experiments found that bacteria could release a myriad of structurally complex metabolites during growth, and many of these metabolites are resistant to degradation (e.g., carboxyl-rich alicyclic molecules, p-amino acids, and numerous structurally-uncharacterized low-molecular-weight compounds; Gruber et al., 2006; Kawasaki & Benner, 2006; Lechtenfeld et al., 2015). The third is through the support of deep-sea free-living microbial communities. Particle-attached bacteria may solubilize substantially more POM than needed for respiration (Smith et al., 1992). The excess hydrolysates support free-living heterotrophs and likely chemoautotrophs as well (e.g., ammonia oxidizers) in the dark ocean (Hansman et al., 2009; Nagata et al., 2000), offering an additional channel for the production of potentially refractory DOM via the microbial carbon pump (Bayer et al., 2019; Jiao et al., 2010). While the magnitude of the last two pathways cannot be constrained with our data, they likely operate in diverse bacterial groups with varied utilization strategies (e.g., external hydrolysis, selfish uptake, passive scavenging; Arnosti et al., 2021). Overall, we suggest that in the context of accumulating evidence that carbon sequestration in the global ocean is largely mediated by bacteria, it may in fact be linked more strongly to the particle-associated bacteria and be more complex than previously thought.

Finally, the observed inverse relationship between bacterial contribution and export production (Figure 3) predicts an increasingly important role of bacteria in the carbon export and sequestration in a warming climate. Ongoing



Figure 4. Conceptual diagram illustrating three potential carbon sequestration pathways mediated by particle-associated bacteria in the ocean. Bacteria attached to particles release exoenzymes that cleave large particles into small organic molecules. These hydrolyzed organics support the growth of attached bacteria, which synthesize new bacterial cell components that are incorporated into the sinking particles (pathway 1). As part of their metabolisms, the attached bacteria release numerous structurally complex exometabolites, including refractory dissolved organic molecules (RDOM) (e.g., carboxyl-rich alicyclic molecules) (pathway 2). The excess hydrolytes taken up by free-living bacteria can be transformed into RDOM via the classic microbial carbon pump (MCP) (pathway 3). Some of the newly synthesized cells and detritus (via pathway 1) as well as RDOM (via pathways 2 and 3) can be exchanged with particles through processes of attachment and detachment.

ocean warming, with associated intensified water stratification and nutrient stress, has been implicated in observed large-scale decreases in phytoplankton size, biomass, and productivity in the surface ocean (Behrenfeld et al., 2006; Daufresne et al., 2009). Export production as a result of these shifts is expected to decrease and be increasingly dominated by small particles (Bopp et al., 2005; Close et al., 2013; Henson et al., 2021; Richardson & Jackson, 2007). Smaller and slower-sinking particles would benefit the growth of particle-attached bacteria (McDonnell et al., 2015; Spilling et al., 2023) and thereby their production of refractory DOM and POM (Figure 4), beyond what would be predicted by export production changes alone. As a result, the production and export of refractory bacterial organic matter are expected to rise as the ocean warms, which in turn may represent a novel negative feedback to rising atmospheric carbon dioxide. As these trends are projected to continue and accelerate (Steinacher et al., 2010), the role of bacteria in marine carbon export and sequestration is expected to become increasingly important.

## **Data Availability Statement**

Research data used in this study are presented in Table S1 in Supporting Information S1.

## References

Amon, R. M. W., Fitznar, H. P., & Benner, R. (2001). Linkages among the bioreactivity, chemical composition, and diagenetic state of marine dissolved organic matter. Limnology & Oceanography, 46(2), 287-297. https://doi.org/10.4319/lo.2001.46.2.0287

Arístegui, J., Gasol, J. M., Duarte, C. M., & Herndl, G. J. (2009). Microbial oceanography of the dark ocean's pelagic realm. Limnology & Oceanography, 54(5), 1501–1529, https://doi.org/10.4319/lo.2009.54.5.1501

Arnosti, C., Wietz, M., Brinkhoff, T., Hehemann, J.-H., Probandt, D., Zeugner, L., & Amann, R. (2021). The biogeochemistry of marine polysaccharides: Sources, inventories, and bacterial drivers of the carbohydrate cycle. Annual Review of Marine Science, 13(1), 81-108. https://doi. org/10.1146/annurev-marine-032020-012810

ing the sediment trap samples. We thank Ronald Benner for his insightful comments on the early versions of the manuscript and for providing the facility for amino acid analysis. We thank the handling editor Dr. Angelicque White and five reviewers for their constructive comments and suggestions. This work was supported by National Natural Science Foundation of China (NSFC) (42106040 to Y.S.), the Fundamental Research Funds for the Central Universities of China (20720210076 to Y.S.), and by the National Science Foundation of the USA (OCE 1635527 to T.G. and M.M). A portion of this work was performed under the auspices of the U.S. Department of Energy (DE-AC52-07NA27344).

conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Cc

19448007, 2023, 11, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022GL102485, Wiley Online Library on [04/07/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/term

- Azam, F., Smith, D. C., & Carlucci, A. F. (1992). Bacterial transformation and transport of organic matter in the Southern California Bight. *Progress in Oceanography*, 30(1–4), 151–166. https://doi.org/10.1016/0079-6611(92)90011-n
- Baumas, C. M., Le Moigne, F. A., Garel, M., Bhairy, N., Guasco, S., Riou, V., et al. (2021). Mesopelagic microbial carbon production correlates with diversity across different marine particle fractions. *The ISME Journal*, 15(6), 1695–1708. https://doi.org/10.1038/s41396-020-00880-z
- Bayer, B., Hansman, R. L., Bittner, M. J., Noriega-Ortega, B. E., Niggemann, J., Dittmar, T., & Herndl, G. J. (2019). Ammonia-oxidizing archaea release a suite of organic compounds potentially fueling prokaryotic heterotrophy in the ocean. *Environmental Microbiology*, 21(11), 4062–4075. https://doi.org/10.1111/1462-2920.14755
- Behrenfeld, M. J., O'Malley, R. T., Siegel, D. A., McClain, C. R., Sarmiento, J. L., Feldman, G. C., et al. (2006). Climate-driven trends in contemporary ocean productivity. *Nature*, 444(7120), 752–755. https://doi.org/10.1038/nature05317
- Benner, R., & Amon, R. M. (2015). The size-reactivity continuum of major bioelements in the ocean. *Annual Review of Marine Science*, 7(1), 185–205. https://doi.org/10.1146/annurev-marine-010213-135126
- Benner, R., & Kaiser, K. (2003). Abundance of amino sugars and peptidoglycan in marine particulate and dissolved organic matter. *Limnology & Oceanography*, 48(1), 118–128. https://doi.org/10.4319/lo.2003.48.1.0118
- Bidle, K. D., & Azam, F. (1999). Accelerated dissolution of diatom silica by marine bacterial assemblages. *Nature*, 397(6719), 508–512. https://doi.org/10.1038/17351
- Boeuf, D., Edwards, B. R., Eppley, J. M., Hu, S. K., Poff, K. E., Romano, A. E., et al. (2019). Biological composition and microbial dynamics of sinking particulate organic matter at abyssal depths in the oligotrophic open ocean. *Proceedings of the National Academy of Sciences*, 116(24), 11824–11832. https://doi.org/10.1073/pnas.1903080116
- Bopp, L., Aumont, O., Cadule, P., Alvain, S., & Gehlen, M. (2005). Response of diatoms distribution to global warming and potential implications: A global model study. *Geophysical Research Letters*, 32(19), L19606. https://doi.org/10.1029/2005gl023653
- Calleja, M. L., Batista, F., Peacock, M., Kudela, R., & McCarthy, M. D. (2013). Changes in compound specific δ<sup>15</sup>N amino acid signatures and D/L ratios in marine dissolved organic matter induced by heterotrophic bacterial reworking. *Marine Chemistry*, 149, 32–44. https://doi. org/10.1016/j.marchem.2012.12.001
- Castro, C. G., Chavez, F. P., Pennington, J. T., Durazo, R., & Collins, C. A. (2018). Temporal variability of downward fluxes of organic carbon off Monterey Bay. Deep Sea Research Part II: Topical Studies in Oceanography, 151, 89–101. https://doi.org/10.1016/j.dsr2.2018.07.001
- Chavez, F. P., Pennington, J. T., Michisaki, R. P., Blum, M., Chavez, G. M., Friederich, J., et al. (2017). Climate variability and change: Response of a coastal ocean ecosystem. *Oceanography*, 30(4), 128–145. https://doi.org/10.5670/oceanog.2017.429
- Cho, B. C., & Azam, F. (1988). Major role of bacteria in biogeochemical fluxes in the ocean's interior. *Nature*, 332(6163), 441–443. https://doi.org/10.1038/332441a0
- Church, M. J., Kyi, E., Hall, R. O., Jr., Karl, D. M., Lindh, M., Nelson, A., & Wear, E. K. (2021). Production and diversity of microorganisms associated with sinking particles in the subtropical North Pacific Ocean. *Limnology & Oceanography*, 66(9), 3255–3270. https://doi.org/10.1002/ lno.11877
- Close, H. G., Shah, S. R., Ingalls, A. E., Diefendorf, A. F., Brodie, E. L., Hansman, R. L., et al. (2013). Export of submicron particulate organic matter to mesopelagic depth in an oligotrophic gyre. *Proceedings of the National Academy of Sciences of the United States of America*, 110(31), 12565–12570. https://doi.org/10.1073/pnas.1217514110
- Cowie, G. L., & Hedges, J. I. (1994). Biochemical indicators of diagenetic alteration in natural organic-matter mixtures. *Nature*, 369(6478), 304–307. https://doi.org/10.1038/369304a0
- Daufresne, M., Lengfellner, K., & Sommer, U. (2009). Global warming benefits the small in aquatic ecosystems. Proceedings of the National Academy of Sciences of the United States of America, 106(31), 12788–12793. https://doi.org/10.1073/pnas.0902080106
- Ducklow, H. W., Hill, S. M., & Gardner, W. D. (1985). Bacterial growth and the decomposition of particulate organic carbon collected in sediment traps. *Continental Shelf Research*, 4(4), 445–464. https://doi.org/10.1016/0278-4343(85)90004-4
- Farnelid, H. M., Turk-Kubo, K. A., & Zehr, J. P. (2016). Identification of associations between bacterioplankton and photosynthetic picoeukaryotes in coastal waters. *Frontiers in Microbiology*, 7(339). https://doi.org/10.3389/fmicb.2016.00339
- Goutx, M., Wakeham, S. G., Lee, C., Duflos, M., Guigue, C., Liu, Z., et al. (2007). Composition and degradation of marine particles with different settling velocities in the northwestern Mediterranean Sea. *Limnology & Oceanography*, 52(4), 1645–1664. https://doi.org/10.4319/ lo.2007.52.4.1645
- Gruber, D. F., Simjouw, J.-P., Seitzinger, S. P., & Taghon, G. L. (2006). Dynamics and characterization of refractory dissolved organic matter produced by a pure bacterial culture in an experimental predator-prey system. *Applied and Environmental Microbiology*, 72(6), 4184–4191. https://doi.org/10.1128/aem.02882-05
- Gupta, L. P., & Kawahata, H. (2002). Impact of ENSO variability on the flux and composition of sinking POM in the western equatorial Pacific Ocean: Amino acids and hexosamines. *Deep Sea Research Part II: Topical Studies in Oceanography*, 49(13–14), 2769–2782. https://doi. org/10.1016/s0967-0645(02)00057-7
- Hansman, R. L., Griffin, S., Watson, J. T., Druffel, E. R., Ingalls, A. E., Pearson, A., & Aluwihare, L. I. (2009). The radiocarbon signature of microorganisms in the mesopelagic ocean. *Proceedings of the National Academy of Sciences of the United States of America*, 106(16), 6513–6518. https://doi.org/10.1073/pnas.0810871106
- Hedges, J. I., Baldock, J. A., Gelinas, Y., Lee, C., Peterson, M., & Wakeham, S. G. (2001). Evidence for non-selective preservation of organic matter in sinking marine particles. *Nature*, 409(6822), 801–804. https://doi.org/10.1038/35057247
- Hedges, J. I., & Keil, R. G. (1995). Sedimentary organic matter preservation: An assessment and speculative synthesis. *Marine Chemistry*, 49(2–3), 81–115. https://doi.org/10.1016/0304-4203(95)00013-h
- Henson, S. A., Cael, B., Allen, S. R., & Dutkiewicz, S. (2021). Future phytoplankton diversity in a changing climate. *Nature Communications*, 12(1), 1–8. https://doi.org/10.1038/s41467-021-25699-w
- Herndl, G. J., & Reinthaler, T. (2013). Microbial control of the dark end of the biological pump. *Nature Geoscience*, 6(9), 718–724. https://doi.org/10.1038/ngeo1921
- Hernes, P. J., Hedges, J. I., Peterson, M. L., Wakeham, S. G., & Lee, C. (1996). Neutral carbohydrate geochemistry of particulate material in the central equatorial Pacific. *Deep-Sea Research Part Ii-Topical Studies in Oceanography*, 43(4–6), 1181–1204. https://doi. org/10.1016/0967-0645(96)00012-4
- Hernes, P. J., Peterson, M. L., Murray, J. W., Wakeham, S. G., Lee, C., & Hedges, J. I. (2001). Particulate carbon and nitrogen fluxes and compositions in the central equatorial Pacific. *Deep-Sea Research Part I-Oceanographic Research Papers*, 48(9), 1999–2023. https://doi.org/10.1016/s0967-0637(00)00115-1
- Ittekkot, V., Deuser, W. G., & Degens, E. T. (1984). Seasonality in the fluxes of sugars, amino-acids, and amino-sugars to the deep ocean Sargasso Sea. *Deep-Sea Research, Part A: Oceanographic Research Papers*, *31*(9), 1057–1069. https://doi.org/10.1016/0198-0149(84)90012-8

- Jiao, N., Herndl, G. J., Hansell, D. A., Benner, R., Kattner, G., Wilhelm, S. W., et al. (2010). Microbial production of recalcitrant dissolved organic matter: Long-term carbon storage in the global ocean. *Nature Reviews Microbiology*, 8(8), 593–599. https://doi.org/10.1038/nrmicro2386
- Johnson, W. M., Longnecker, K., Kido Soule, M. C., Arnold, W. A., Bhatia, M. P., Hallam, S. J., et al. (2020). Metabolite composition of sinking particles differs from surface suspended particles across a latitudinal transect in the South Atlantic. *Limnology & Oceanography*, 65(1), 111–127. https://doi.org/10.1002/lno.11255
- Kaiser, K., & Benner, R. (2005). Hydrolysis-induced racemization of amino acids. *Limnology and Oceanography: Methods*, 3(8), 318–325. https://doi.org/10.4319/lom.2005.3.318
- Kaiser, K., & Benner, R. (2008). Major bacterial contribution to the ocean reservoir of detrital organic carbon and nitrogen. Limnology & Oceanography, 53(1), 99–112. https://doi.org/10.4319/lo.2008.53.1.0099
- Kaiser, K., & Benner, R. (2009). Biochemical composition and size distribution of organic matter at the Pacific and Atlantic time-series stations. *Marine Chemistry*, 113(1–2), 63–77. https://doi.org/10.1016/j.marchem.2008.12.004
- Kawasaki, N., & Benner, R. (2006). Bacterial release of dissolved organic matter during cell growth and decline: Molecular origin and composition. *Limnology & Oceanography*, 51(5), 2170–2180. https://doi.org/10.4319/lo.2006.51.5.2170
- Kawasaki, N., Sohrin, R., Ogawa, H., Nagata, T., & Benner, R. J. A. M. E. (2011). Bacterial carbon content and the living and detrial bacterial contributions to suspended particulate organic carbon in the North Pacific Ocean. *Aquatic Microbial Ecology*, 62(2), 165–176. https://doi. org/10.3354/ame01462
- Lechtenfeld, O. J., Herkorn, N., Shen, Y., Witt, M., & Benner, R. (2015). Marine sequestration of carbon in bacterial metabolites. *Nature Communications*, 6(1), 6711. https://doi.org/10.1038/ncomms7711
- Lee, C., Wakeham, S., & Arnosti, C. (2004). Particulate organic matter in the sea: The composition conundrum. Ambio, 33(8), 565–575. https:// doi.org/10.1579/0044-7447-33.8.565
- Lee, C., Wakeham, S. G., & Hedges, J. I. (2000). Composition and flux of particulate amino acids and chloropigments in equatorial Pacific seawater and sediments. *Deep-Sea Research Part I-Oceanographic Research Papers*, 47(8), 1535–1568. https://doi.org/10.1016/ s0967-0637(99)00116-8
- Lehmann, M. F., Carstens, D., Deek, A., McCarthy, M., Schubert, C. J., & Zopfi, J. (2020). Amino acid and amino sugar compositional changes during in vitro degradation of algal organic matter indicate rapid bacterial re-synthesis. *Geochimica et Cosmochimica Acta*, 283, 67–84. https:// doi.org/10.1016/j.gca.2020.05.025
- McCarthy, M. D., Benner, R., Lee, C., & Fogel, M. L. (2007). Amino acid nitrogen isotopic fractionation patterns as indicators of heterotrophy in plankton, particulate, and dissolved organic matter. *Geochimica et Cosmochimica Acta*, 71(19), 4727–4744. https://doi.org/10.1016/j. gca.2007.06.061
- McDonnell, A. M., Boyd, P. W., & Buesseler, K. O. (2015). Effects of sinking velocities and microbial respiration rates on the attenuation of particulate carbon fluxes through the mesopelagic zone. *Global Biogeochemical Cycles*, 29(2), 175–193. https://doi.org/10.1002/2014gb004935
- Mestre, M., Ruiz-González, C., Logares, R., Duarte, C. M., Gasol, J. M., & Sala, M. M. (2018). Sinking particles promote vertical connectivity in the ocean microbiome. *Proceedings of the National Academy of Sciences of the United States of America*, 115(29), E6799–E6807. https:// doi.org/10.1073/pnas.1802470115
- Minor, E. C., Wakeham, S. G., & Lee, C. (2003). Changes in the molecular-level characteristics of sinking marine particles with water column depth. Geochimica et Cosmochimica Acta, 67(22), 4277–4288. https://doi.org/10.1016/s0016-7037(03)00263-1
- Nagata, T., Fukuda, H., Fukuda, R., & Koike, I. (2000). Bacterioplankton distribution and production in deep Pacific waters: Large-scale geographic variations and possible coupling with sinking particle fluxes. *Limnology & Oceanography*, 45(2), 426–435. https://doi.org/10.4319/ lo.2000.45.2.0426
- Nagata, T., Meon, B., & Kirchman, D. L. (2003). Microbial degradation of peptidoglycan in seawater. *Limnology & Oceanography*, 48(2), 745–754. https://doi.org/10.4319/10.2003.48.2.0745
- Nguyen, T. T., Zakem, E. J., Ebrahimi, A., Schwartzman, J., Caglar, T., Amarnath, K., et al. (2022). Microbes contribute to setting the ocean carbon flux by altering the fate of sinking particulates. *Nature Communications*, 13(1), 1657. https://doi.org/10.1038/s41467-022-29297-2
- Ogawa, H., Amagai, Y., Koike, I., Kaiser, K., & Benner, R. (2001). Production of refractory dissolved organic matter by bacteria. *Science*, 292(5518), 917–920. https://doi.org/10.1126/science.1057627
- Preston, C. M., Durkin, C. A., & Yamahara, K. M. (2020). DNA metabarcoding reveals organisms contributing to particulate matter flux to abyssal depths in the North East Pacific ocean. Deep Sea Research Part II: Topical Studies in Oceanography, 173, 104708. https://doi. org/10.1016/j.dsr2.2019.104708
- Richardson, T. L., & Jackson, G. A. (2007). Small phytoplankton and carbon export from the surface ocean. Science, 315(5813), 838–840. https:// doi.org/10.1126/science.1133471
- Robinson, R. S., Kienast, M., Luiza Albuquerque, A., Altabet, M., Contreras, S., De Pol Holz, R., et al. (2012). A review of nitrogen isotopic alteration in marine sediments. *Paleoceanography*, 27(4), PA4203. https://doi.org/10.1029/2012pa002321
- Roland, L. A., McCarthy, M. D., & Guilderson, T. (2008). Sources of molecularly uncharacterized organic carbon in sinking particles from three ocean basins: A coupled Δ14C and δ13C approach. *Marine Chemistry*, 111(3–4), 199–213. https://doi.org/10.1016/j.marchem.2008.05.010
- Salter, I., Kemp, A. E. S., Lampitt, R. S., & Gledhill, M. (2010). The association between biogenic and inorganic minerals and the amino acid composition of settling particles. *Limnology & Oceanography*, 55(5), 2207–2218. https://doi.org/10.4319/lo.2010.55.5.2207

Schleifer, K. H., & Kandler, O. (1972). Peptidoglycan types of bacterial cell walls and their taxonomic implications. *Bacteriological Reviews*, 36(4), 407–477. https://doi.org/10.1128/br.36.4.407-477.1972

- Shen, Y., & Benner, R. (2020). Molecular properties are a primary control on the microbial utilization of dissolved organic matter in the ocean. *Limnology & Oceanography*, 65(5), 1061–1071. https://doi.org/10.1002/lno.11369
- Shen, Y., Benner, R., Murray, A. E., Gimpel, C., Mitchell, B. G., Weiss, E. L., & Reiss, C. (2017). Bioavailable dissolved organic matter and biological hot spots during austral winter in Antarctic waters. *Journal of Geophysical Research: Oceans*, 122(1), 508–520. https://doi. org/10.1002/2016jc012301
- Shen, Y., Guilderson, T. P., Sherwood, O. A., Castro, C. G., Chavez, F. P., & McCarthy, M. D. (2021). Amino acid \u03b313C and \u03b315N patterns from sediment trap time series and deep-sea corals: Implications for biogeochemical and ecological reconstructions in paleoarchives. *Geochimica et Cosmochimica Acta*, 297, 288–307. https://doi.org/10.1016/j.gca.2020.12.012
- Simon, M., Alldredge, A. L., & Azam, F. (1990). Bacterial carbon dynamics on marine snow. *Marine Ecology Progress Series*, 65, 205–211. https://doi.org/10.3354/meps065205
- Smith, D. C., Simon, M., Alldredge, A. L., & Azam, F. (1992). Intense hydrolytic enzyme-activity on marine aggregates and implications for rapid particle dissolution. *Nature*, 359(6391), 139–142. https://doi.org/10.1038/359139a0
- Spilling, K., Heinemann, M., Vanharanta, M., Baumann, M., Noche-Ferreira, A., Suessle, P., & Riebesell, U. (2023). Respiration rate scales inversely with sinking speed of settling marine aggregates. *PLoS One*, 18(3), e0282294. https://doi.org/10.1371/journal.pone.0282294

- Steinacher, M., Joos, F., Frölicher, T., Bopp, L., Cadule, P., Cocco, V., et al. (2010). Projected 21st century decrease in marine productivity: A multi-model analysis. *Biogeosciences*, 7(3), 979–1005. https://doi.org/10.5194/bg-7-979-2010
- Suzuki, M. T., Preston, C. M., Chavez, F. P., & DeLong, E. F. (2001). Quantitative mapping of bacterioplankton populations in seawater: Field tests across an upwelling plume in Monterey Bay. Aquatic Microbial Ecology, 24(2), 117–127. https://doi.org/10.3354/ame024117
- Tremblay, L., Caparros, J., Leblanc, K., & Obernosterer, I. (2015). Origin and fate of particulate and dissolved organic matter in a naturally iron-fertilized region of the Southern Ocean. *Biogeosciences*, 12(2), 607–621. https://doi.org/10.5194/bg-12-607-2015
- Turley, C., & Mackie, P. (1994). Biogeochemical significance of attached and free-living bacteria and the flux of particles in the NE Atlantic Ocean. Marine Ecology Progress Series, 115, 191–203. https://doi.org/10.3354/meps115191
- Turner, J. T. (2015). Zooplankton fecal pellets, marine snow, phytodetritus and the ocean's biological pump. Progress in Oceanography, 130, 205–248. https://doi.org/10.1016/j.pocean.2014.08.005
- Waite, A. M., Safi, K. A., Hall, J. A., & Nodder, S. D. (2000). Mass sedimentation of picoplankton embedded in organic aggregates. *Limnology & Oceanography*, 45(1), 87–97. https://doi.org/10.4319/lo.2000.45.1.0087
- Wakeham, S. G., & Lee, C. (1989). Organic geochemistry of particulate matter in the ocean: The role of particles in oceanic sedimentary cycles. Organic Geochemistry, 14(1), 83–96. https://doi.org/10.1016/0146-6380(89)90022-3
- Wakeham, S. G., & Lee, C. (2019). Limits of our knowledge, Part 2: Selected frontiers in marine organic biogeochemistry. *Marine Chemistry*, 212, 16–46. https://doi.org/10.1016/j.marchem.2019.02.005
- Wakeham, S. G., Lee, C., Hedges, J. I., Hernes, P. J., & Peterson, M. L. (1997). Molecular indicators of diagenetic status in marine organic matter. Geochimica et Cosmochimica Acta, 61(24), 5363–5369. https://doi.org/10.1016/s0016-7037(97)00312-8
- Wakeham, S. G., Pease, T. K., & Benner, R. (2003). Hydroxy fatty acids in marine dissolved organic matter as indicators of bacterial membrane material. Organic Geochemistry, 34(6), 857–868. https://doi.org/10.1016/s0146-6380(02)00189-4
- Zhao, Z., Baltar, F., & Herndl, G. J. (2020). Linking extracellular enzymes to phylogeny indicates a predominantly particle-associated lifestyle of deep-sea prokaryotes. *Science Advances*, 6(16), eaaz4354. https://doi.org/10.1126/sciadv.aaz4354

## **References From the Supporting Information**

- Chavez, F. P. (1996). Forcing and biological impact of onset of the 1992 El Niño in central California. Geophysical Research Letters, 23(3), 265–268. https://doi.org/10.1029/96gl00017
- Kim, T.-H., Kim, G., Shen, Y., & Benner, R. (2017). Strong linkages between surface and deep-water dissolved organic matter in the East/Japan Sea. *Biogeosciences*, 14(9), 2561–2570. https://doi.org/10.5194/bg-14-2561-2017
- McCarthy, M. D., Hedges, J. I., & Benner, R. (1998). Major bacterial contribution to marine dissolved organic nitrogen. Science, 281(5374), 231–234. https://doi.org/10.1126/science.281.5374.231
- Paerl, R. W., Turk, K. A., Beinart, R. A., Chavez, F. P., & Zehr, J. P. (2012). Seasonal change in the abundance of Synechococcus and multiple distinct phylotypes in Monterey Bay determined by rbcL and narB quantitative PCR. *Environmental Microbiology*, 14(3), 580–593. https:// doi.org/10.1111/j.1462-2920.2011.02594.x
- Pilskaln, C. H., Lehmann, C., Paduan, J. B., & Silver, M. W. (1998). Spatial and temporal dynamics in marine aggregate abundance, sinking rate and flux: Monterey Bay, central California. Deep Sea Research Part II: Topical Studies in Oceanography, 45(8), 1803–1837. https://doi. org/10.1016/s0967-0645(98)80018-0

#### **Erratum**

In the originally published version of this article, Figure 4 was missing a blue wave at the top that indicates surface ocean, due to a typesetting error. The figure has been corrected, and this may be considered the authoritative version of record.